Author's response to reviews

Title: Analysis of clinical uncertainties by health professionals and patients: an example from mental health

Authors:

Keith Lloyd (k.r.lloyd@swansea.ac.uk)
Matteo Cella (m.cella@swansea.ac.uk)
Michael Tanenblatt (mtan@us.ibm.com)
Anni Coden (anni@us.ibm.com)

Version: 10 Date: 6 July 2009

Author's response to reviews:

Dear editor,

the acknowledgements have been updated as requested to read: "KL conceived the study and participated in its design and coordination. KL and MC carried out the original data collection. All authors conceived and carried out the data analysis and interpretation. All authors read and approved the final manuscript."

Reference 10 has been presented in the same style and with the same abbreviations as when it has been cited elsewhere in Biomed Central. There is no volume number. It now reads "Huang X, Lin J, Demner-Fushman D: Evaluation of PICO as a knowledge representation for clinical questions. AMIA Annu Symp Proc. 2006, 359-63."

Thank you